Author’s response to reviews

Title: Hepatholithiasis with biliary ascariasis - A case report

Authors:

Dr Kirtikumar S Pilankar (kirtikumarp@yahoo.com)
Dr Anjali D Amarapurkar (anjali_1963@hotmail.com)
Rajeev M Joshi (drrajeevjoshi@yahoo.com)
Tilakdas S Shetty (tsshetty@rediff.com)
Amit S Khitani (amit khithani@yahoo.com)
Vipul V Chemburkar (vipulchem@indiatimes.com)

Version: 8 Date: 12 Oct 2003

Respected Madam/Sir,

I am submitting herewith revised manuscript with certain suggested changes as mentoined by Mohammad S Khuru and Luca Antonio Aldrighetti.
1. consent has been faxed to your number.
2. legends to the figures are written on separate sheet.
3. minor gramatical mistakes and corrections in references have been done.
kindly let me know whether you received consent or not.
Abovementioned changes have been made in manuscript. Kindly do the needful.
Thanking you in anticipation.

Dr. Kirtikumar S. Pilankar